Re: Submission of a Revised manuscript to the WORLD JOURNAL OF GASTROENTEROLOGY

Thank you very much for giving us the opportunity to revise our invited review (Manuscript number: 77462)

We would be grateful for the consideration of our revised manuscript “Comments on "Development and validation of a prediction model"” for publication in the World Journal of Gastroenterology.moderately severe and severe acute pancreatitis in pregnancy” (by Qunying Yang, Jianwen Hu)

The authors have read and complied with author guidelines, and they all have seen and approved this manuscript for publication. None of the authors had a conflict of interest to disclose concerning this manuscript.

We are grateful to the Editors for their precious contributions and comments. We have revised our manuscript accordingly for grammar, style, structure and we hope that you will now find it suitable for publication in the World Journal of Gastroenterology. In case of final acceptance, we agree to make this manuscript open-access.

The changes in the manuscript are identified in track change mode. Below you can find a point-by-point reply to the reviewers. We used red to denote revised or inserted text.

Thank you for your precious time.
We are looking forward to receiving your decision in due time.

Correspondence author
Jianwen Hu, PhD, Professor, Department of Gastroenterology, Dongyang People's Hospital, No. 60 Wuning West Road, Dongyang 322100, Zhejiang Province, China.
18367908196@163.com
Reply to Reviewers

1. Reviewer #1: Please upload the full manuscript. In general: it's a good paper and the subject of the manuscript is applicable and useful. Title: the title properly explains the purpose and objective of the article. Abstract: abstract contains an appropriate summary for the article. Language used in the abstract is easy to read and understand. There are no suggestions for improvement. Grammar: Need some revision. (Check the paper comments). Please provide the following information in the paper:
1. Conflict of Interest
2. Source of Funding
Finally, this was an appealing article, in its current state it adds much new insightful information to the field. Therefore, I accept that paper to be published in your journal.

Thanks to the reviewer for the appreciation of our paper. We have made modifications according to your recommendations. We have sent the revised manuscript to a professional English language editing company and provided a new language certificate along with the manuscript. We have provided the following information in the paper:
1. Conflict-of-interest statement: Each author stated that there is no conflict of interest.
2. The study was not funded or sponsored.

2. Reviewer #2: The authors have raised some valid points regarding the published articles. However, there are some grammatical and typographical errors which must be addressed.

Thanks to the reviewer for the appreciation of our paper. We have sent the revised manuscript to a professional English language editing company for further language polishing. We have updated the format of our manuscript according to the Guidelines and Requirements for Manuscript Revision and the Format for Manuscript Revision.